# Voluntary Quits: Do Works Councils Matter? An Analysis of the Reform of the German Works Constitution Act 2001 \*

#### Julian B. Adam<sup>1</sup>

<sup>1</sup>Friedrich-Alexander-University Erlangen-Nuremberg Department of Economics Chair of Empirical Economics E-mail: julian.b.adam@fau.de

**Abstract.** Most of the literature on the effects of German works councils does not deal with the issue of potential endogeneity of works council existence. Exploiting exogenous variation in works council authority stemming from a 2001 reform of the German Works Constitution Act, I apply a regression differencein-difference using establishment panel data. I find that increasing works council size and the introduction of one full-time councilor causally reduces the number of voluntary quits by about 30 percent. This decline is driven entirely by collective voice effects and there is no evidence for monopoly effects in place. Similar to the findings of previous research, the effect is significant only in establishments which are subject to a collective agreement. The results suggest that the effectiveness of works councils either heavily relies on the support of unions, or that works councils mainly serve as a guardian of collective agreements.

#### JEL-Classification: J51, J53, J63

**Keywords**: works councils, codetermination, industrial relations, quits, collective bargaining

#### 1. Introduction

Due to search, vacancy and training costs, voluntary quits of employees can entail substantial costs to employers as well as to employees encountering a loss in employerspecific human capital or the threat of unemployment spells in between two jobs. Information asymmetries or a public good problem of enforcing good conditions of work can render separations arising from voluntary quits inefficient, which is why the determinants and causes of voluntary quits have for a long time been attracting the interest of research in economics. Bryant and Allen (2013) emphasize that, contrary to a common belief, remuneration is not one of the most decisive factors of voluntary quits, but rather overall job satisfaction, organizational commitment, the relationship with supervisors and advancement opportunities. The German Institute for Employment Research reports that 84 percent of employees in Germany suffered from deadline pressures, information overload, physical strains or unpleasant work environments at the workplace (Wolter et al.

<sup>\*</sup>I am grateful for valuable comments and helpful suggestions by two anonymous referees, Regina Riphahn, Daniel Kühnle, Thomas Bauer and Markus Hertwig, the participants of the 14th IWH/IAB Workshop on Labor Market Policy and of the IAB 2017 International Workshop on Establishment Panel Analyses.

2016) and that these conditions increased the likelihood of job change considerations among employees. Especially when the labor market is tight, skilled labor is scarce or during high demand on the product market, the costs of turnover are high and employers have a strong interest in retaining employees. Dietz et al. (2013) find that quits are most prevalent in small and medium-sized establishments. They argue that training opportunities, larger internal labor markets and workplace representation and codetermination lead to relatively fewer quits in larger establishments.

There are various forms of institutions governing the relationship between employers and employees that influence the conditions of work and pay and that are suspected to influence conditions of work and pay, potentially affecting job satisfaction and quit behavior of employees.

This paper considers the reform of the German Works Constitution Act of 2001 which, among others, intended to strengthen the authority of works councils by raising the compulsory minimum number of councilors as well as the number of councilors which are granted paid release from work for certain establishment sizes only. Applying a differencein-difference research design for the identification of causal effects and using data from the German Establishment Panel from 1998 through 2004, the change in size thresholds determining works council size is shown to have a significantly negative and causal effect on voluntary quits. Disentangling voice and monopoly effects, I find that the reduction in quits is a likely consequence of improved working conditions retaining workers and not a consequence of increased rent redistribution triggered by the enlargement of works councils. I argue, that the findings with respect to the effects of increasing works councils' power are likely to resemble the effects of works council existence and can at least qualitatively be interpreted in that way. In line with previous literature, significant effects are shown to exist only in establishments subject to collective agreements. The contribution to the empirical literature on the effects works councils is twofold. The paper is among the first to analyze the effect of one particular aspect, the enlargement of works councils, of the German 2001 reform of the Works Constitution Act. Moreover, the identification strategy takes into account the potential endogeneity of works council status and accordingly allows a causal interpretation of the findings, which is rare in the existing literature.

After decades of research on the effects of works councils, starting with rather negative evaluations of FitzRoy and Kraft (1985) and FitzRoy and Kraft (1987), in recent decades the pendulum has swung towards a more positive assessment of the economic effects of works councils. Notwithstanding that the empirical literature remains overall inconclusive, what most studies have in common is that they do not allow for a causal interpretation of the effects of workplace representation and codetermination, which is the novelty in this study.

The remainder of the paper is as follows: section 2 presents the institutional background of the German system of workplace representation and codetermination as well as the reform of the Works Constitution Act of 2001. Section 3 discusses theoretical effects of works councils on quits and provides an overview of the findings from previous empirical research. Section 4 describes the data and identification strategy. The estimation results as well as results from various robustness checks are presented in section 5 and section 6 provides a discussion of the results and points at potential limitations of the research.

Section 7 concludes.

#### 2. Works Councils in Germany: Institutional Background

#### 2.1. Employee Representation and Codetermination in Germany

When it comes to the effects of codetermination, the case of Germany is frequently analyzed. There are some distinctive features that make it commonly being considered the archetype of a system of workplace representation and codetermination. Although many southern European and some central and eastern European countries also feature dual systems of employee representation with unions and workplace organizations, the extent to which works councils have codetermination rights in addition to information and consultation rights is very limited and there typically exists a power imbalance in favor of the labor unions. The situation in Germany is characterized by unions, rather than works councils, engaging in collective bargaining, by the works councils' obligation not to engage in industrial action and their far-reaching rights and engagement in a variety of management decisions. In international comparison it appears to be the most comprehensive<sup>1</sup> and formalized system while strongly guided by the principle of cooperation. This seeming contradiction and the complementarity of unions and works councils has for decades been attracting interest in its functioning and the economic consequences. In the following, the institutional background of codetermination and the 2001 reform of the Works Constitution Act will be described.

The German system of worker representation is characterized by its dual, or complementary, structure. At the industry level, unions represent employees' interests by engaging in industrial disputes and participating in the collective bargaining process, which, among others, determines minimum wages for a wide range of sectors. In addition to unions at the industry level, works councils can be founded in every establishment with at least five employees of the age over 18 years, and if at least three of them are employed for a minimum of six months. Once established, there exists no statutory procedure to abolish a works council. The members of the council, called councilors, are elected every four years by the workforce and their number is based on the total number of employees in the establishment and determined by law. Additionally, the Works Constitution Act prescribes a minimum number of councilors who are granted paid leave from work<sup>2</sup>, who work for the council on a full-time basis, which also depends on the number of employed persons in an establishment. All costs of council election, initiation and operation are borne by the employer. The tasks of the works councils comprise the initiation of measures serving the interests of employees<sup>3</sup> and the enforcement of regulations laid down in the labor code or collective agreements<sup>4</sup>, preventing workplace discrimination, dealing with environmental aspects of the establishment and training measures as well as participating in personnel decisions such as hirings, firings or the transformation of employment contracts<sup>5</sup>. In practice, it serves as a means of individual and collective communication

<sup>&</sup>lt;sup>1</sup>For a more contextual interpretation of German works councils as a role model for European codetermination, see Addison (2009).

<sup>&</sup>lt;sup>2</sup>In the following, these will also be referred to as released councilors.

<sup>&</sup>lt;sup>3</sup>Betriebsverfassungsgesetz (2001), 80, sect. 1(2)

<sup>&</sup>lt;sup>4</sup>Betriebsverfassungsgesetz (2001), 80, sect. 1(1)

<sup>&</sup>lt;sup>5</sup>Betriebsverfassungsgesetz (2001), 80, sect. 1(7) and sect. 1(9)

between employees and employers.<sup>6</sup>

The influence of works councils stems from the comprehensive statutory rights, which comprise information rights, consultation rights and codetermination rights. Works councils are granted comprehensive information rights which means that, with few exceptions, all information that is required for the work of the councils has to be provided either on a regular basis or upon request. These, among others, can comprise data on employees, information on hirings and firings, trainings, labor and environmental protection and works alteration. Consultation rights grant the works councils the opportunity of consulting with the employer. The codetermination rights are the most powerful tool works councils have in representing the employees' interests towards the management. The most important of these are the actual right of codetermination, the right of objection and the right to refuse consent. It should be noted that councils in practice have two ways of exerting power and influence: they can object on a case-by-case basis, depending on the respective issue, or they can arbitrarily oppose the management's plans in order to achieve concessions in future affairs or unrelated matters as well as simultaneously negotiate a variety of different aspects or measures in so-called bundling agreements.

Despite the frequently noticed decline in the coverage of works councils and collective agreements in the recent decades, in 2000, according to Schnabel and Kohaut (2003) still about 71% (55%) of all employees in West (East) Germany worked in establishments subject to a collective agreement and 54% (47%) in establishments with a works council, as Addison et al. (2003) state.

## 2.2. The Reform of the Works Constitution Act of 2001

The existence of work councils in Germany dates back to the Weimar Republic, when the first law governing the foundation of work councils in establishments with more than 20 employees was passed after a period of ongoing industrial conflict in 1920. Under the Nazi regime all forms of organization of workers, unions as well as work councils, were prohibited. After WWII, the Works Constitution Act (*Betriebsverfassungsgesetz, BetrVG*) was passed in 1952. It comprised the regulations regarding the initiation of councils, procedural regulations and rights and duties of works councils and employers. After a reform in 1972, the last major amendment of the Works Constitution Act took place in 2001<sup>7</sup> and came into effect in July 2001. The reform intended to strengthen the role of works councils and to adapt workplace codetermination to a modern labor market and works environment as well as to the emergence and proliferation of irregular forms of employment like agency work.

One component of the reform was the simplification of the council election procedure by lowering the requirements for the initiation of a works meeting which can decide on the foundation of a works council. This was intended to encourage the formation of councils especially in small and medium-sized establishments in which works councils rarely exist. The foundation of company-wide works councils embracing multiple establishment works councils was also simplified. Furthermore, the right to vote in the council elections was extended to agency workers with a tenure of at least three months in the respec-

<sup>&</sup>lt;sup>6</sup>For a more detailed description of the role, rights and responsibilities of works councils in Germany,see Addison (2009), chapter 2.

<sup>&</sup>lt;sup>7</sup>For the full text, see Bundesgesetzblatt (2001) (in German language).

tive establishment.<sup>8</sup> In order to prevent the excessive use of service contracts which hire workers for specific tasks on a fixed-term basis and are not subject to social security payments, the obligatory consent of the works council to such hirings was introduced. The law also intended to stipulate worker participation in the council work by forcing works councils to take up issues that are supported by at least five percent of the workforce. Secondly, the tasks of the works councils were complemented by matters of environmetal protection, promotion of work-life balance, and prevention of racism. Furthermore, the reform act obliged the management to consider job security measures proposed by the works councils. Another important change concerned the composition of works councils: the amended law mandates that the minority sex in the establishment needs to be represented at least proportionally to its share in the workforce in councils with at least three councilors.

Finally and most important for this research, the amendment did not only strengthen the power of works councils qualitatively by giving it more responsibilities and rights, but also quantitatively by increasing the total number of councilors as well as the number of councilors which need to be released from work. This change did not apply to all establishments, but only to establishments with certain sizes of the workforce and the arbitrariness of the threshold values gives rise to the identification strategy described in section 4. Table 1 presents the number of councilors and releases before and after the reform.<sup>9</sup> Whereas the size of works councils increased by two in the affected establishments, the number of paid releases was raised by one for selected establishment sizes.<sup>10</sup> Establishments with 201 to 300 employees were not only affected by the increase in the number of councilors, but they were also required to grant paid leave to one councilor for the first time, whereas before the reform the threshold for the first release was 300 employees. Figure 1 and 2 show the prescribed minimum number of councilors and released councilors depending on the number of regularly employed persons before and after the reform. The dark bars reflect the situation before the amendment in 2001 and the gray shaded parts reflect the increases in the number of councilors and councilors with paid leave.

Whereas the increase of works councils was supposed to happen after the forthcoming universal council elections in 2002, additional releases were supposed to happen immediately after the amendment came into effect in July 2001. Although the parliamentary debates took place only in April 2001 after the Federal Government had published the draft bill, reforming the Works Constitution Act has been a recurring issue in the political discussion during the years preceding the amendment. The intensive discussion was initially triggered by a publication of a proposed draft bill of the Works Constitution Act by the German Trade Union Confederation (*Deutscher Gewerkschaftsbund - DGB*), see DGB (1998). Following this initiative, other confederated union organizations published competing drafts and thus, fueled the debate on modernizing the works constitution.

<sup>&</sup>lt;sup>8</sup>The reason for this provision is that these workers are employed legally by temporary work agencies whose potential works councils cannot exert influence with respect to the working conditions in the hiring establishments.

<sup>&</sup>lt;sup>9</sup>It also shows the numbers stemming from a proposal of the German Trade Union Confederation of 1998. The comparison suggests that the law was modeled closely after this proposal.

<sup>&</sup>lt;sup>10</sup>The reason why the number of councilors was increased by an even number is due to the fact that works councils always consist of an odd number of councilors in order to avoid ties in simple majority votings.

#### 3. Effects of Works Councils on Quits: Theory and Evidence

Works councils represent all employees in an establishment and can theoretically act as monopolists of labor by raising wages above the competitive level. In the German institutional setting, where unions and works councils are complementary institutions with different responsibilities, unions engage in industry level collective bargaining and wage agreements, if applicable. However, it is conceivable that works councils generate rents for workers in a similar manner as unions do. Addison et al. (2001) find that the existence of works councils is associated with higher wages and lower profitability.<sup>11</sup> However, they also explicitly acknowledge that the councils' rent-seeking behavior is constrained by the collective bargaining process at the sectoral level. Using the IAB Establishment Panel, Bellmann and Kohaut (1999) find a significantly positive relationship between works council status and per-capita wages in East Germany only. Jirjahn (2003) finds a positive association of works council existence and wages, the effect being stronger for establishments not subject to collective agreements. It should be noted that not considering collective agreements may lead to spurious results and conclusions, since the existence of works councils and collective agreements are correlated and accordingly the effect of the latter might be erroneously attributed to works councils. Despite the fact that most theoretical considerations suggest a positive association between works councils and wages, the opposite may also occur, for instance during downturns, when wage cuts are necessary to secure the continuation of the firm and employment. In such a situation, works councils may lower the workers' resistance to wage cuts by means of mediation.<sup>12</sup>

As Freeman and Medoff (1979) argue, the range of potential effects of worker representation is not limited to monopoly effects. In the context of employee-employer communication and potentially beneficial effects of worker representation on firms, Freeman and Medoff (1979) take up the phrase of collective voice or institutional response (view), of which the term voice dates back to Hirschman (1970), who used it in the context of customer (dis-)satisfaction. In the collective voice framework, unions or works councils provide workers with a means of expressing needs, dissatisfaction and preferences and may prevent workers from choosing their exit option.<sup>13</sup>

Consequently, the term voice effect refers to changes in establishment and employment outcomes related to communicating and enforcing workers' non-monetary preferences and proposals for organizational restructuring, whereas monopoly effect denotes a distribution of rents from employers to employees effected by workplace codetermination. These are the definitions which most closely follow the meaning in Freeman and Medoff (1979). It is also necessary to regard the two concepts in terms of economic efficiency. In case of monopoly effects only, works councils are inefficient since they lead to higher wages and consequently too low levels of employment. When both effects are at work, the

<sup>&</sup>lt;sup>11</sup>Higher wages alone are not sufficient to find the monopoly view confirmed, since they could be the effect of a productivity increase potentially induced by works councils, which needs to be controlled for.

<sup>&</sup>lt;sup>12</sup>Especially when the primary objective of works councils is securing jobs as opposed to generating additional rents for the employees, this is likely to be the case.

<sup>&</sup>lt;sup>13</sup>Taking into account that working and workplace conditions constitute a public good and are potentially subject to the classical free-rider problem, employees may not individually communicate their needs and preferences towards the management. As in the classical free-rider problem, the individual reluctance to openly engage in an improvement of working conditions will result in a suboptimal provision of these. Moreover, individual critique or suggestions might be perceived as a risk to the individual, increasing the probability of dismissal or other negative consequences.

effect on efficiency depends on the relative size of monopoly and voice effect. If there are only voice effects in place, works councils are plausibly pareto-improving, by mitigating the negative consequences of market failure due to the public good problem of working conditions and of the employment contract being "necessarily and inherently incomplete as well as unaviodably controversial" (Frick 1996: 409).

A reduction in quits results in lower search and vacancy costs which increases productivity and does not require firm-specific training of new workers which would temporarily reduce productivity. To the degree that the collective voice view applies, Freeman and Medoff (1979) argue that the detrimental effects of works councils on profitability caused by the monopoly status of works councils or unions may be partially, fully or more than fully offset by the works councils' role as collective voice, preventing voluntary quits and fostering productivity. If the two views and effects are considered jointly, it remains ambiguous whether works councils are beneficial or harmful to firm performance and this theoretical ambiguity accordingly calls for an empirical assessment. The effect on quits is less ambigious, since monopoly and voice effects are both expected to reduce voluntary quits.

Sadowski et al. (1995) find that works councils are associated with a 2.4 percentage points lower rate of voluntary quits, which is of a similar order of magnitude as the 1.6 percentage points Frick (1996) reports. Since they do not report the average quit rate in their sample, the relative size of the effect cannot be evaluated. Nevertheless, even for an unusually high quit rate of up to 10 percent, the effect appears to be substantial in magnitude. Using data from the Socio-economic Panel, Grund et al. (2016) find that works councils are associated with a 1.2 percentage point lower quit rate, which corresponds to a 27 percent difference in voluntary guits between establishments with and without a works council, respectively. Pfeifer (2007) reports that the existence of a works council is associated with 20 percent fewer voluntary quits, whereas the difference is 30 percent when works councils and collective agreements co-exist. Boockmann and Steffes (2010) show that works councils are associated with longer job duration of male workers, with a decrease in the job exit hazard of more than 20 percent. Applying a matching approach using data from the IAB Establishment Panel, Addison et al. (2004), do, however, not find a statistically significant causal effect of works councils on quits.<sup>14</sup> Hirsch et al. (2010) find that works councils are associated with a separation rate which is 1.5 percentage lower than in the absence of works councils, which in turn corresponds to 13 percent. Disentangling monopoly and voice effects, they confirm the existence of the monopoly effect for all types of workers, whereas the voice effect is heterogenous across different kinds of workers and only significant for male workers with low tenure. Both Addison et al. (2004) and Hirsch et al. (2010) use the separation rate as a measure for voluntary quits. The separation rate also includes dismissals, exits to retirement and deaths and therefore, the effect on voluntary quits is likely to be different from the effect reported. Pfeifer (2011a) provides some additional descriptive evidence in favor of the monopoly view of works councils. Distinguishing between works councils with different degrees

<sup>&</sup>lt;sup>14</sup>They compare establishments in which a works council has been founded with establishments without a works council for the years 1996-2000. Since the initiation of a works council is a relatively rare event, their analysis is based on approximately 30 pairs of observations and the insignificance of all coefficient estimates might be caused by the small sample size. The insignificant estimate of the effect of works councils on the quit rate is positive, though.

of consensuality, statistically significant rent-sharing effects are reported. It should be noted that, except for the the paper of Addison et al. (2004), the designs of the studies mentioned do not allow for a causal interpretation of the effects. Since works councils are not automatic, i.e. they need to be initiated by the workforce, their existence cannot be considered exogenous. To the degree that the initiation of a works council is correlated with establishment-specific unobservable characteristics that simultaneously affect voluntary quits, this endogeneity problem may lead to a biased estimation. Furthermore, the examined literature looks at the effect of the mere existence of works councils, but not the size, composition or number of full-time councilors that potentially determine the assertiveness and power of councils and the size of the effects accordingly.

From the theoretical considerations and the empirical findings, the following hypotheses with respect to voluntary quits of workers can be derived:

Hypothesis 1: The release of one councilor reduces the number of voluntary quits in establishments.

This hypothesis is based on the monopoly and collective voice view of works councils. Both rent-generating and working condition-improving councils potentially reduce the number of quits. It is assumed that the release of councilors increases the assertiveness and the authority of works councils, particularly in case of the first release. A negative causal effect on voluntary quits does, however, not imply that both effects are at work or that both effects are of equal magnitude, from which the second hypothesis is derived:

Hypothesis 2: In the presence of collective agreements concluded by labor unions, the negative effect of works councils on voluntary quits is mainly driven by the voice effect.

As mentioned before, in Germany collective agreements are negotiated by labor unions and trade unions. These usually comprise sectoral minimum wages, working time regulations and general working conditions. In this institutional setting, the influence of works councils on wages is likely to be limited and the effect of works councils on voluntary quits is then caused mainly by an improvement of individual- and establishment-, but not of industry-specific working conditions. Notwithstanding, not all firms are subject to collective agreements and the relative size of the effects may differ between firms with and without a collective agreements.

The complementary structure of German worker representation and workplace codetermination further raises interest in the effect of the coexistence of works councils and collective bargaining. Pfeifer (2011b) mentions several conceivable ways of how works councils and membership in a trade union or collective agreements can interact. Firstly, unions may directly support the work of works councils by providing advice or sharing other resources. Secondly, works councils could improve the enforcement of collective agreements. With respect to rent-seeking activities, it is often argued that collective agreements at the industry level reduce the degree of distributional conflicts at the establishment level. The industry level of collective bargaining and the limited bargaining power of works councils could lead to a reduction of works council rent-seeking and a greater involvement in productivity-enhancing activities. From this point of view, the interaction of works councils and industry-level collective agreements could reinforce the effects and in particular increase productivity. A number of studies seem to empirically confirm this view with respect to productivity and profits. With regard to fluctuation, Frick and Möller (2003) find that the effect of works councils on personnel turnover is negative throughout, but larger in establishments subject to a collective agreement. Looking at voluntary quits, Pfeifer (2011b) finds a negative effect of works councils exclusively in establishments subject to a collective agreement.

Hypothesis 3: The effect of works councils on voluntary quits through the monopoly effect is stronger in establishments which are not subject to collective agreements.

This hypothesis follows directly from the consideration that with a collective agreement, most of the conflict about the distribution of rents is carried out at the industry, but not at the establishment level. The willingness or capability of employers to grant rents to employees may also vary considerably conditional upon collective agreements. Taking into account that collective agreements do not only govern the remuneration, but also govern general working conditions, the following hypothesis shall be tested:

Hypothesis 4: The effect of works councils on voluntary quits through the voice effect is weaker in establishments which are not subject to collective agreements.

The same argument as with respect to hypothesis 3 could apply and works councils be considered as constrained to matters that are not regulated elsewhere, e.g. by collective agreements. To the degree that collective agreements contain employee-friendly regulations regarding non-pecuniary conditions of work, the voice function of works councils might be obsolete. However, taking into account time and resource constraints of works councils, it is also plausible that the absence of wage negotiations at the establishment level allows for more engagement with respect to its voice function. Accordingly, the effect via the voice channel is expected to be more prevalent in establishments with a collective agreement, which is supported by the analysis of the general effects of works councils as found in Frick and Möller (2003) concerning fluctuations and in Pfeifer (2011a) concerning productivity, as well as by the reasoning of Freeman and Lazear (1995). Furthermore, the absence of the distributional conflict at the establishment level may improve the relationship between works councils and management and facilitate the improvement of working conditions through voice.

The multiplicity of plausible arguments with respect to voice and monopoly effects of works councils and its potential interaction with collective agreements once again emphasizes the necessity of an empirical analysis of the causal effects. The causal identification strategy and the data used for estimation are presented in the following section.

## 4. Empirical analysis

## **4.1. Data** <sup>15</sup>

I am using the IAB Establishment Panel which is, to the best of my knowledge, the only representative and large establishment dataset for Germany. Since 1993 in West and 1996

<sup>&</sup>lt;sup>15</sup>The data analyzed stems from the IAB Establishment Panel, waves 1998-2004. The data was accessed on-site at the Research Data Centre of the Federal Employment Agency at the Institute for Employment Research. The research is registered under the project number fdz1348.

in East Germany, respectively, it is annually conducted for the Institute of Employment Research (IAB) of the German Federal Employment Agency and the reference day for reporting is June 30. It consists of 8000 to 16000 observations of establishments per year which are randomly selected from the Federal Employment Agency's register of employment. The sample is disproportionately stratified across industries, federal states and establishment sizes. Large establishments are overrepresented, which makes it particularly feasible for the analysis of the policy change with respect to medium-sized and large establishments.<sup>16</sup> The Establishment Panel mainly focuses on topics of employment, personnel policies and training, participation in active labor market policies, the overall economic situation, investments, innovations and technological change. In Germany, it is the only representative large-scale employer dataset containing information on the works council status. The sample analyzed consists of all establishments with 151-250 employees with a persistent works council from 1998-2004 and no change in the collective agreement status during this period. Table 2 reports descriptive statistics for the whole sample as well as separately for the chosen treatment and control group. The average number of regularly employed persons is around 199 and the average number of quits in one year is around 2.7, which corresponds to a quit rate of 1.4 percent. Of all 2638 establishments in the sample, 83 percent are subject to a collective agreement and this share does not differ across the treatment and control group.<sup>17</sup>

#### 4.2. Identification strategy

The empirical analysis of the causal effects of works councils bears some methodological pitfalls, one being the potential endogeneity of works council status. Although works councils in Germany are mandatory, they need to be initiated by the workforce and as such they do not exist in all establishments. The concern is that the initiation and consequently the existence of works councils might be a function of establishment-specific unobservable characteristics that also affect the outcome of interest, voluntary quits. In that case, the problem of endogeneity does not allow for a causal interpretation. Methods to identify the causal economic effects of works councils comprise matching techniques, as well as the exploitation of quasi-experimental settings.

The discontinuous changes in the threshold number of employees determining the number of councilors and councilors with paid release from work provide a suitable setting for a quasi-experimental study design. I argue that the effect of increasing works councils leads to an increase of assertiveness and power and shows qualitatively similar effects as the existence of works councils. This is not an inconceivable assumption, considering that councilors often perform their work after regular working time and are subject to time and knowledge or experience constraints. The assumption is more likely to hold for the release of the first councilor from work than for merely increasing the number of councilors or released councilors. Whereas the latter may lead to coordination problems or increased heterogeneity within the councils, potentially jeopardizing assertiveness and consentaneity, the initial release of one full-time councilor is expected to unambiguously strengthen the stand of the works council.

I apply a difference-in-difference (DiD) estimation strategy to analyze the causal effect

<sup>&</sup>lt;sup>16</sup>For a detailed methodological description, see Kölling (2000) and Fischer et al. (2009).

<sup>&</sup>lt;sup>17</sup>The share of establishment with collective agreement is somewhat higher than in the population, probably since the sample is restricted to establishments with works councils.

of increasing works councils on voluntary quits. Establishments with 201-250 employees serve as the treatment group. The treatment comprises the first release of a full-time councilor from work<sup>18</sup> in July 2001, as well as an increase in the number of councilors from seven to nine, which happened in the context of the next council elections in the first half of 2002. Establishments with 151-200 employees serve as the control group. As can be seen also from Table 1, these establishments were not affected neither by an increase nor by a release. Other changes induced by the reform, as described in section 2.2 do not threaten the identification, since no other provision affected these two groups in a different manner. That nothing else that could have affected voluntary quits changed at the time of the reform differently for these two groups of establishments, is one of the prerequisites for the validity of the interpretation of the difference-in-difference (DiD) method. The difference in the differential development of voluntary guits between treated and untreated establishments can then be interpreted as the causal effect of the treatment on the outcome, voluntary quits. Another necessary assumption for a correct parametric estimation in the context of this DiD identification strategy is that in the absence of the regulatory change, the number of quits would have followed the same time trend for both groups. The common trend assumption is fundamentally untestable, but the graphical evidence in Figure 3 roughly supports the claim of common trends in the unconditional outcome variable for the selected treatment and control group. Moreover, there is no reasonable explanation for a potential violation of the common trend assumption for the treatment and control group determined by an arbitrary size threshold. The effect uncovered by DiD is the average treatment effect on the treated.

Lechner (2011) lists additional assumptions that are required in order to render the DiD identification strategy internally valid. Besides the assumption that the treatment does not affect the general equilibrium or untreated establishments other than directly by means of the treatment, which Manski (2013) labeled the individualistic treatment response assumption, exogeneity of the treatment is required. Exogeneity is usually assumed in the context of legislative changes and if the possibility or incentives for self-selection into treatment and control group are not affected by the change. One way to rule out selfselection is by comparing transitions between treatment and control group before and after the time of the treatment. Such a comparison does not reveal an increase or decrease in the number of threshold transitions after the reform. Furthermore, self-selection into the control group, e.g. by dismissals and deferred or suspended hirings, is most likely to occur in case of establishments with a number of employees close to the threshold.<sup>19</sup> Taking this into account, the robustness checks presented in chapter 5 contain the results from an analysis for a subsample, that does not comprise establishments close to the threshold of 200 employees. Finally, it needs to be ruled out that there is an effect of the treatment on the pre-treatment outcome. If this assumption is violated the DiD method is still applicable, however, some treatment effect might be observed prior to the actual treatment, potentially as an anticipation of expected better working conditions or higher wages.

<sup>&</sup>lt;sup>18</sup>The reform also introduced the possibility of releasing several councilors from work on a part-time basis. Unfortunately, the data at hand does not contain information on actual full-time or part-time releases and there is no information on the extent to which this is used in practice. However, it is plausible that the release of two part-time councilors has similar effects as the release of one full-time councilor.

<sup>&</sup>lt;sup>19</sup>This is due to the cost of deviating from the previously optimal number of employees. The closer the number of employees is to the treatment threshold of 200 employees, the more likely will the anticipated expected cost of an increased works council exceed the cost of deviating from the original stock of labor.

The basic specifications of the regression equation are

$$y_{it} = \mathbf{X}_{it}\boldsymbol{\beta} + \delta_1 treat_i + \delta_2 post_t + \gamma treat_i \times post_t + \eta_i + \epsilon_{it}$$
(1)

$$y_{it} = \mathbf{X}_{it}\boldsymbol{\beta} + \delta_1 treat_i + \sum_{t=1998}^{2000} \zeta_t year_t + \sum_{t=2002}^{2004} \zeta_t year_t + \sum_{t=1998}^{2000} \gamma_t year_t \times treat_i + \sum_{t=2002}^{2004} \gamma_t year_t \times treat_i + \eta_i + \epsilon_{it}$$

$$(2)$$

where  $y_{it}$  is the number of voluntary quits of establishment *i* in year *t*,  $\mathbf{X}_{it}$  contains establishment characteristics such as business volume, the total number of employees and industry dummies. In order to account for time-invariant unobserved heterogeneity, the model is estimated with establishment-fixed effects  $\eta_i$ . In equation (1), the coefficient of interest, the treatment effect, is given by  $\gamma$ . Equation (2), instead of using a dummy variable for the time after the reform,  $post_t$ , contains treatment-year interactions in order to analyze the evolution of the treatment effect during the years after the reform, which not only allows for testing the assumption of no effect on the pre-treatment outcome, but also for the study of the treatment effect over time.

#### 5. Results

Column (1) of Table 3 reports the results from the estimation of the baseline specification. Control variables in the regression comprise the natural log of sales, total number of employees in the establishment a dummy indicating whether the establishment is bound by a collective agreement<sup>20</sup>, a dummy for West Germany and nine industry dummys. In the affected establishments, granting paid leave to one councilor and the increased council size reduced the number of voluntary quits by about 0.9. Taking into account the mean of voluntary quits in the treatment group, this amounts to a reduction of 30 percent. The effect is statistically significant at the 1%-level and its magnitude is similar to the one found by Boockmann and Steffes (2010), larger than Hirsch et al. (2010) and presumedly smaller than the result of Sadowski et al. (1995). Note that the above mentioned studies do not study the effect of the 2001 Works Constitution Act reform, but analyze the association between quits or fluctuation with the general existence of works councils, which even more stresses the substantial magnitude of the effect.

Following the approach of Hirsch et al. (2010), column (2) of Table 3 reports the results from a regression including a measure of productivity and wages, the natural log of value added per worker and the wage sum per capita for the month June, in order to account for potential indirect effects of the monopoly mechanism. The authors argue that if productivity and wages are controlled for, the reported coefficient of the treatment effect reflects the mere voice effect and the presence of monopoly effects would lead to a smaller coefficient than in column (1). Since the effect is not smaller than in the baseline model, there is no evidence for monopoly effects leading to a reduction of voluntary quits. Columns (3)

<sup>&</sup>lt;sup>20</sup>The collective agreement can either be a sectoral agreement or a company wide agreement.

and (4) show the results from the augmented regression distinguishing the treatment effect by collective agreement status. Whereas there is no significant treatment effect found for establishments without a collective agreement, the treatment reduced voluntary quits by approximately one in establishments which are subject to a collective agreement and the effect is significant at the 1%-level in all specifications.<sup>21</sup> In neither establishments with, nor without a collective agreement there is evidence for monopoly effects. This finding is in line with Frick and Möller (2003) and Pfeifer (2011b) who found that the effects of works councils are more pronounced in case of the co-existence of works councils and collective agreements.

Analyzing the course of the treatment effect in more detail, Table 4 presents estimates from a regression that contains treatment-year interactions instead of the interaction of the treatment indicator with a post-reform dummy. Before the reform, all treatment effects are insignificant, which is crucial for the validity of the causal identification of the effects. On average, the treatment reduced the number of quits by 1.1 in 2002 compared to 2001 and the estimated coefficient is statistically significant at the 5%-level. The difference between the effect in 2003 and 2001 is 1.63 and statistically highly significant. The difference between the estimates of the treatment effect in column 5 and 6 is statistically not significant, which suggests that only voice effects are at work. The treatment effect in 2004 is insignificant in all specifications, which provides evidence for a merely temporary effect on voluntary quits. A similar result is described by Grund et al. (2016), who find that the formation of works councils initially increases job satisfaction, but this effect vanishes within not more than five years after formation.<sup>22</sup> If the dependent variable is the quit ratio instead of the number of quits,<sup>23</sup> the results are fairly similar: the treatment reduces the quit ratio by 0.3 percentage points, which corresponds to a reduction of about 25 percent in the mean quit rate of 1.3 percent in the group of treated establishments.<sup>24</sup>

Several robustness checks have been carried out, the results of which shall be discussed in the remainder of the section. One potential source of error could occur when employers misreport the number of employees in the Establishment Panel. Instead of the exact number, respondents may report rounded numbers, which would pose a threat to the chosen identification strategy that is relying on the sharp threshold of 200 employees. The size distribution of establishments appears to be uniform and there is no evidence for bunching at the threshold. Another, more serious, threat is self-selection into control or treatment group. Self-selection into the treatment group is unlikely to occur, since establishments can voluntarily grant works councils a higher number of members or paid leave. Selection into the control group by dismissals or suspended hirings, however, could cause the effects to be overestimated. In order to avoid this potential pitfall, the models have been estimated with a reduced sample exempting all establishments with a number of employ-ees close to the threshold of 200 employees. Tables 5 and 6 report the results from the estimation based on the sample of establishments with 150-195 employees and 205-250

<sup>&</sup>lt;sup>21</sup>The treatment effect for establishments with a collective agreement is the sum of the coefficients of the variables Treat  $\times$  Post and Treat  $\times$  Post  $\times$  Coll. Agr. Statistics and p-values from an F-test on joint significance are reported in the table notes.

<sup>&</sup>lt;sup>22</sup>Due to restrictions in the data they use, they cannot observe the exact duration of the reversion of the effect.

<sup>&</sup>lt;sup>23</sup>That is defined as the ratio of the number of quits in a given year to employment.

<sup>&</sup>lt;sup>24</sup>The results are reported in Tables 12 and 13.

employees. In addition, Tables 7 and 8 show the results from the estimation based on a sample of establishments with 150-190 and 210-250 employees. The estimated coefficients are significant and of a similar magnitude as before, which suggests the validity of the results.<sup>25 26</sup> Moreover, a change in the number of transitions between treatment and control group after the report could suggest self-selection induced by the reform. There is no such changed pattern being observed in the data. Table 9 shows the results of a placebo test of specifications (1)-(4) with the treatment being predated to the year 2000. The treatment effect is insignificant, as can be seen from the respective values of the F-statistic.<sup>27</sup>

Since the dependent variable takes non-negative integers and about 30 percent of observations in the sample display zero voluntary quits, in addition to the fixed effects OLS regression, a fixed effects Poisson regression model for count data has been estimated.<sup>28</sup> <sup>29</sup> The effects suggested by the OLS and Poisson fixed effects regressions are virtually the same<sup>30</sup> and also the levels of statistical significance remain almost unchanged. Another issue is the number of employees, which is the variable that assigns establishments to the treatment. According to the Works Constitution Act, this number contains all regularly employed persons in an establishment<sup>31</sup>, but working owners are explicitly exempted. The results are insensitive to the small deviations in the size of the workforce caused by regarding working owners. None of the conducted robustness checks puts the validity of the findings into question.

#### 6. Caveats and Discussion

One recurrent concern in the context of granting paid release to members of works councils is the danger of potential non-compliance with the threshold provisions.<sup>32</sup>Apart from some reportedly confidential anecdotal evidence which Koller et al. (2008) mention and

<sup>&</sup>lt;sup>25</sup>The size thresholds for these subsamples have been chosen in order not to overly drastically reduce the sample size. Apart from that, any self-selection was likely to occur as a result of anticipated cost of increased works councils exceeding costs of deviating from the previous number of employees, which becomes the less likely, the greater the deviation is.

<sup>&</sup>lt;sup>26</sup>The relevant establishment size refers to the number of employees and does not regard working owners. In order to account for small discrepancies caused by counting working owners as employees when reporting the number of employed persons, I have subtracted these from the size of the workforce in an additional robustness check. The results are robust with repect to this modification.

<sup>&</sup>lt;sup>27</sup>The same analysis has been conducted for the observation period 1998-2002, and another analysis has been conducted with the reform predated to 1999 for the observation period 1998-2000. In all these analyses, the treatment effects are insignificant. The results of all robustness checks not reported in the appendix are available from the author upon request.

<sup>&</sup>lt;sup>28</sup>Wooldridge (2010) shows, that the violation of the central assumption of the Poisson regression model, the mean-variance equality, does not lead to an inconsistent estimator for and that inference based on robust standard errors is valid.

<sup>&</sup>lt;sup>29</sup>The obtained coefficients for the treatment effects are reported in Tables 10 and 11. The detailed results are available from the author upon request.

<sup>&</sup>lt;sup>30</sup>In order to be able to directly compare the magnitude of the coefficients of the treatment effect, the coefficients need to be multiplied by the conditional mean of the dependent variable. For the sake of an easier interpretation, only OLS results are being reported and discussed in the text of the paper.

<sup>&</sup>lt;sup>31</sup>This threshold regulation is based on the number of persons, and not based on the number of full-time equivalents.

<sup>&</sup>lt;sup>32</sup>This regards establishments with a smaller number of council members and releases as well as establishments voluntarily exceeding the minimum requirements specified in the Works Constitution Act.

which is difficult to quantify and to use in empirical research, Mohrenweiser and Backes-Gellner (2010) provide survey-based evidence for a considerable degree of non-compliance of establishments with the provisions of the Works Constitution Act with respect to releases. The authors base their conclusions on a works council survey of the Institute for Research on Small- and Medium-sized Establishments.<sup>33</sup> Taking into account the extremely low response rate of less than seven percent, the risk of selection into response requires caution when interpreting the results.<sup>34</sup> The seeming extent of non-compliance might resemble that responding "good" employers in establishments below the size threshold are more likely to voluntarily grant paid release or works councils may relinquish additional rights granted by the reform act. In that case, the actual degree of non-compliance would be lower than suggested in that survey and emphasized by the authors. Unfortunately, with the data from the IAB Establishment Panel, the extent of non-compliance cannot be quantified and evaluated in this paper. Even if the IAB Establishment Panel contained information on the actual number of released councilors, it remains questionable whether employers would reveal unlawful practices.

Besides intentional non-compliance, there can be other circumstances preventing employers or works councils from complying with the changed works council size and release requirements. Behrens (2003) argues that the adaption of the new regulations may require some time and accordingly compliance may not be immediate. This view could be supported by the significance and magnitude of the coefficients of the treatment-year interactions for the years 2002 and 2003 reported in Tables 4, 6, 8, and 11. However, it is not possible to unambiguously state whether the evolvement of the effects over time is due to delayed compliance or to a delayed effect itself.

The remainder of this section will discuss the findings reported in section 5. Although mostly in line with previous literature, little is known about the determinants of the relative size of voice and monopoly effects and the persistence of the effects of works councils on voluntary quits. One potential explanation for the relative magnitude of the effects reported, especially in comparison with the findings of e.g. Boockmann and Steffes (2010) or Hirsch et al. (2010), who are analyzing the consequences of works council status, is the endogeneity of works council existence. The effect in previous studies is likely to be underestimated if works council existence was negatively correlated with unobserved establishment characteristics that are associated with voluntary quits, for instance an uncooperative management or a bad work environment. Due to these factors, the formation of works councils could be inhibited or workers council is initiated, as Lücking (2006) describes. That could imply that works councils are more likely to exist in good establishments, which could explain the greater magnitude of the effect in this causal analysis, where the treatment as opposed to works council status, is assumed to be exogenous.<sup>35</sup>

Another remaining question refers to the lack of evidence for monopoly effects in this study. It can plausibly be argued that in establishments with a collective agreement distri-

<sup>&</sup>lt;sup>33</sup>Institut für Mittelstandsforschung (author's translation)

<sup>&</sup>lt;sup>34</sup>Although this is a works council survey, it was sent directly to employers, who were asked to forward it to the works councils, and there might be some extent of employer selection in forwarding or not forwarding the surveys.

<sup>&</sup>lt;sup>35</sup>However, the opposite argument of works councils being more likely to be founded in bad establishments is conceivable, as well.

butional conflicts are not or only to a lesser extent carried out at the establishment level. Accordingly, the finding that significant treatment effects are observed in establishments with collective agreements only provides additional support to the evidence of exclusive voice effects in this group of establishments, as stated in Hypothesis 2. However, this argument does not explain why no monopoly effects are observed in establishments without collective bargaining. Freeman and Lazear (1995) hypothesize that for this reason works councils in establishments with a collective agreement have more capacities left to deal with non-monetary aspects of the employer-employee relationship and thus, their effect with respect to productivity and other outcomes might be more pronounced, which is the finding in most of the literature. This reasoning could explain why the voice effect was smaller in establishments without collective agreements, but it cannot explain the simultaneous absence of voice and monopoly effects in establishments without collective agreement coverage. The absence of monopoly effects in general is likely to be the result from the dual nature of the German system of employee representation and codetermination that does not assign wage and remuneration bargaining to works councils, but explicitly to unions. Acknowledging that works councils may still behave monopolistically to some extent and attempt to increase remuneration, one can say this behavior was more likely to occur in case of no collective agreement and potentially prevents those works councils from engagement with respect to other aspects like working conditions. This could explain the missing evidence of an effect of the treatment on voluntary quits by the voice mechanism in establishments without a collective agreement.

Alternatively, the actual role of works councils could differ from the classical view that works councils autonomously and actively act as a codetermination organ. If instead the councils acted as the guardian of treaties, in this case collective agreements, the observed difference in effects between establishments with and without collective agreements, respectively, would reflect the effect of a higher degree of enforcement of provisions stemming from collective bargaining. Finally, there may be considerable interactions between works councils and unions. Whether, as well as to which degree works councils and unions are in contact or collaborating, might depend on collective agreement coverage and the significantly negative treatment effect could consequently be the result of union support to works councils, as Pfeifer (2011b) argues. This support by unions can comprise financial means, the provision of expertise and experience as well as the establishment of contacts among works councils and accordingly increase the efficacy of works councils. However, the interactions of the treatment effect with collective agreement status should be interpreted with caution. As has been argued, the causal effect of works councils is assessed by looking at exogenous variation in the number of released councilors. Collective agreements, though, may not be fully exogenous in this analysis.

The insignificant coefficients of the treatment-year interactions for the year 2004 are somewhat puzzling. As can be seen from Table 5, the estimated treatment coefficient is still negative and of considerable magnitude, but statistically not significant. Accordingly, the initially very strong effect appears to decrease over time and there is a reversion of quits to the original level of 2001. This finding is in line with Grund et al. (2013), who find that the increase in employees' job satisfaction following the introduction of works councils is of a temporary nature and vanishes within six years after the introduction, which they attribute to unmet expectations of the workforce in the works council. If the job satisfaction or quitting behavior was driven by expectations rather than by actual

actions and achievements of the works council, it would be likely to observe an effect already before the actual change in the Works Constitution Act in 2001, since the reform act has been discussed at length and with increasing intensity since the German Trade Union Confederation (DGB) proposed an amendment in 1998, after which the actual reform act was modeled. The question under which conditions and for which reasons the effects are temporary, is beyond the scope of this paper and is left for future research.

# 7. Conclusion

In the last decades, researchers have been extensively dealing with the evaluation of the effects of German works councils on the performance of firms and personnel fluctuation. Although the evidence is mixed, especially contemporary studies do not find negative impacts and often find positive effects of workplace codetermination. A caveat of these previous analyses is the neglection of the potential endogeneity of the works council status, which may bias the results and lead to spurious conclusions. This paper is among the first to analyze the effect of authority gains of works councils on voluntary quits. Using a reform of the German Works Constitution Act of 2001 which affected establishments of certain sizes by raising the minimum number of councilors and councilors with paid release, the difference-in-difference identification strategy allows the findings to be interpreted causally. It shows that the causal effect of strengthening works councils is negative, statistically significant and of considerable magnitude. In establishments with 201-250 employees, the legislative change with respect to the size of works councils and paid leave of councilors on average reduced the number of quits by 0.9, which corresponds to a reduction by 30 percent. The magnitude of the effect is in line with previous findings of the effect of works council existence.

Disentangling voice and monopoly channels of codetermination, I provide evidence that the effect on quits is exclusively driven by the collective voice function of works councils and not by productivity-constant wage increases as a result of monopoly effects. Comparing the effect across establishments with and without binding collective agreements reveals that establishments not subject to a collective agreement do not seem to be affected. This is also in line with previous findings of more pronounced effects of codetermination in case of co-existence with collective bargaining.

There is some evidence for a temporary nature of the effect of the legislative change on voluntary quits. Despite the fact that previous research obtained similar results, not much is known about the dynamics and potential impermanence of the effects of codetermination, yet. One limitation of the analysis is that it focuses on the effects of one particular aspect of the Works Constitution Act, namely the enlargement of works councils, and it remains uncertain to which degree the effects found resemble the effects of works council existence. Since the initial paid release of one full-time councilor is expected to substantially increase the power and to slacken time as well as coordination and knowledge constraints of councils, it is plausible that the results qualitatively apply to works council status, but also by size and authority of works councils, is of particular relevance to employers considering a voluntary release of a councilor or opposing works council power, to employees in different institutional settings faced with the problem of employee retention, or during times of skills shortages.

Although frequently observed, little is known about the underlying mechanisms in which works councils, unions and collective bargaining interact in the German dual system of worker codetermination. Understanding these interactions remains both a task and a challenge for future industrial relations research.

#### References

- Addison, J. T. (2009). *The Economics of Codetermination. Lessons From the German Experience.* Palgrave MacMillan.
- Addison, J. T., Bellmann, L., Schnabel, C., and Wagner, J. (2003). German Works Councils Old and New: Incidence, Coverage and Determinants. *Schmollers Jahrbuch*, 123(3):339–358.
- Addison, J. T., Bellmann, L., Schnabel, C., and Wagner, J. (2004). The Reform of the German Works Constitution Act: A Critical Assessment. *Industrial Relations*, 43(2):392– 420.
- Addison, J. T., Schnabel, C., and Wagner, J. (2001). Works Councils in Germany: Their Effects on Firm Performance. *Oxford Economic Papers*, 53:659–94.
- Behrens, M. (2003). Wie gut kehren neue Besen? Das neue Betriebsverfassungsgesetz in der Praxis. Mitteilungen, WSI.
- Bellmann, L. and Kohaut, S. (1999). Betriebliche Lohnbestimmung in ost- und westdeutschen Betrieben. In Bellmann, L., Kohaut, S., and Lahner, M., editors, Zur Entwicklung von Lohn und Beschäftigung auf der Basis von Betriebs- und Unternehmensdaten, pages 9–25. Institute for Employment Research, Nuremberg.
- Betriebsverfassungsgesetz (2001). Edition of September 25, 2001. *In: Bundesgesetzblatt*, I(I):2518.
- Boockmann, B. and Steffes, S. (2010). Workers, Firms, or Institutions: What Determines Job Duration for Male Employees in Germany? *Industrial and Labor Relations Review*, 64(1):109–27.
- Bryant, P. C. and Allen, D. G. (2013). Compensation, Benefits and Employee Turnover: HR Strategies for Retaining Talent. *Compensation and Benefits Review*, 45(3):171–75.
- Bundesgesetzblatt (2001). Gesetz zur Reform des Betriebsverfassungsgesetzes (BetrVerf-Reformgesetz) vom 23. Juli 2001. Federal Ministry of Justice and Consumer Protection.
- DGB (1998). Novellierungsvorschläge des DGB zum Betriebsverfassungsgesetz 1972. Technical report, Deutscher Gewerkschaftsbund.
- Dietz, M., Kubis, A., Leber, U., Müller, A., and Stegmaier, J. (2013). Kleine und mittlere Betriebe im Wettbewerb um Fachkräfte. Kurzbericht, IAB.
- Fischer, G., Janik, F., Müller, D., and Schmucker, A. (2009). European Data Watch. The IAB Establishment Panel - Things Users Should Know. *Schmollers Jahrbuch*, 129:133–48.
- FitzRoy, F. R. and Kraft, K. (1985). Unionization, Wages, and Efficiency: Theories and Evidence from the US and Germany. *Kyklos*, 38(38):537–554.
- FitzRoy, F. R. and Kraft, K. (1987). Efficiency and internal organization: Works councils in west german firms. *Economica*, 54(216):493–504.
- Freeman, R. B. (1976). Individual Mobility and Union Voice in the Labor Market. *American Economic Review*, 66(2):361–68.
- Freeman, R. B. and Lazear, E. (1995). An Economic Analysis of Works Councils. In Rogers, J. and Streeck, W., editors, *Works Councils: Consultation, Representation*,

and Cooperation in Industrial Relations, pages 27–52. University of Chicago Press, Chicago.

- Freeman, R. B. and Medoff, J. L. (1979). The Two Faces of Unionism. *The Public Interest*, 57:69–93.
- Frick, B. (1996). Co-determination and Personnel Turnover: The German Experience. *Labour Economics*, 10(2):407–30.
- Frick, B. and Möller, I. (2003). Mandated Works Councils and Firm Performance: Labor Productivity and Personnel Turnover in German Establishments. *Schmollers Jahrbuch*, 123:423–54.
- Grund, C., Martin, J., and Schmitt, A. (2013). Works Councils, Wages and Job Satisfaction. *Applied Economics*, 45(3):299–310.
- Grund, C., Martin, J., and Schmitt, A. (2016). Works Councils, Quits and Dismissals in Germany. *German Journal of Human Resource Management*, 30(1):53–75.
- Hirsch, B., Schank, T., and Schnabel, C. (2010). Works Councils and Separations: Voice, Monopoly, and Insurance Effects. *Industrial Relations*, 49(4):566–92.
- Hirschman, A. O. (1970). Exit, Voice, and Loyalty. Responses to Decline in Forms, Organizations, and States. Harvard University Press, Cambridge.
- Jirjahn, U. (2003). Betriebsräte, Tarifverträge und betriebliches Lohnniveau. *Mitteilungen aus der Arbeitsmarkt- und Berufsforschung*, 36(4):649–60.
- Koller, L., Schnabel, C., and Wagner, J. (2008). Freistellung von Betriebsräten Eine Beschäftigungsbremse? *Zeitschrift für Arbeitsmarktforschung*, 2 and 3:305–26.
- Kölling, A. (2000). The IAB Establishment Panel. *Journal of Applied Social Science Studies*, 120(2):291–300.
- Lechner, M. (2011). The Estimation of Causal Effects by Difference-in-Difference Methods. *Foundations and Trends in Econometrics*, 4(3):165–224.
- Lücking, S. (2006). Labor Relations in Enterprises Without Works Council in Germany. 14th International Industrial Relations Association World Congress. *September*, 29:2017.
- Manski, C. F. (2013). Identification of Treatment Response With Social Interactions. *The Econometrics Journal*, 16(1):1–23.
- Mohrenweiser, J. and Backes-Gellner, U. (2010). Die Wirkung des Betriebsverfassungsgesetzes am Beispiel der Freistellung von Betriebsräten - ein Beitrag zur Rechtstatsachenforschung. *Jahrbücher für Nationalökonomie und Statistik*, 4:420–35.
- Pfeifer, C. (2007). Betriebsräte, Tarifverträge und freiwillige Kündigungen von Arbeitnehmern. WSI Mitteilungen, 2:63–9.
- Pfeifer, C. (2011a). The Heterogenous Economic Consequences of Works Council Relations. *Schmollers Jahrbuch*, 131:59–71.
- Pfeifer, C. (2011b). Works Councils, Union Bargaining and Quits in German Firms. *Economic and Industrial Democracy*, 32(2):243–60.
- Sadowski, D., Backes-Gellner, U., and Frick, B. (1995). Works Councils: Barriers or Boosts for the Competitiveness of German Firms? *British Journal of Industrial Relations*, 33(3):493–513.
- Schnabel, C. and Kohaut, S. (2003). Tarifverträge nein danke? ausmaß und einflussfaktoren der tarifbindung west- und ostdeutscher betriebe. Jahrbücher für Nationalökonomie und Statistik, 223(3):312–331.
- Wolter, S., Broszeit, S., Frodermann, C., Grunau, P., and Bellmann, L. (2016). Befragung von Betrieben und Beschäftigten. Mehr Zufriedenheit in Betrieben mit guter Personalpolitik. Kurzbericht, IAB.

Wooldridge, J. (2010). Econometric Analysis of Cross Section and Panel Data. *The MIT Press, Cambridge*, (2nd ed.).

# **Appendix: Figures and Tables**



Figure 1. Number of councilors before and after the 2001 reform, by number of regularly employed persons in the establishment.



Figure 2. Number of released councilors before and after the 2001 reform, by number of regularly employed persons in the establishment.

Ξ,	
200	
рц	
2 a	
197	
of '	
Ţ	
A L	
itio	
titu	
suc	
ŭ	
rks	
Š	
ę	
ing	
ord	
õ	
รั	
Silo	
nn	
õ	
ed	
eas	
ē	
pu	
S a	
iloi	
nnc	
õ	
of	
ber	
m	
ч	
ate	
bne	>
Ř	Vel
<u> </u>	9Cti
able	spe
Ĕ	Ð

Diff. no. of released councilors	0	0	0	0	0	1	0	0	1	0	0	0	1	0	0	1	1	1	1	cs Constitution
Diff. no. of councilors	0	0	0	7	0	0	0	0	7	0	0	0	7	0	0	0	7	0	0	ording to Work
released councilors WCA 2001	0	0	0	0	0	1	1	1*	5	0	7	5*	c	ю	3*	4	4	S	5	councilors acc
released councilors DGB proposal	0	0	0	0	0		1	6	7	0	0	ω	С	ю	4	4	4	S	5	s and released
released councilors WCA 1972	0	0	0	0	0	0	1	1		7	6	6	7	ю	ю	ω	ω	4	4	er of councilor
councilors WCA 2001	1	ω	S	7	7	6	9*	11	11*	11*	13*	13*	13*	15*	15*	$17^{*}$	$17^{*}$	19*	19*	inimum numbe
councilors DGB proposal	1	n	S	7	7	6	11	11	13	13	15	15	17	17	19	21	23	23	25	he mandated m
councilors WCA 1972	1	ω	S	S	7	7	6	6	6	11	11	11	11	15	15	15	15	19	19	s table shows tl
No. of employees	5-20	21-50	51-100	101-150	151-200	201-300	301-400	401-500	501-600	601-700	701-800	801-900	901-1000	1001-1200	1201-1500	1501-1800	1801-2000	2001-2200	2201-2500	Notes: Thi:

Act 1972 and Works Constitution Act 2001. Gray shaded cells indicate changes by the reform legislation. Asterisks denote deviations in the legislation from the 1998 proposal of the German Trade Union Confederation (DGB).

				Mean and std. d	lev.		u	
Variable	Minimum	Maximum	Sample	Ctrl	Treat.	Sample	Ctrl.	Treat.
West	0	1	0.669	0.650	0.690	2638	1398	1240
			(0.471)	(0.477)	(0.463)			
Employees	151	250	198.843	175.220	225.477	2638	1398	1240
			(28.965)	(14.775)	(14.137)			
Sales	14735	36800000000	91800000	60100000	127000000	2638	1398	1240
			(78300000)	(13200000)	(113000000)			
Value added p.c.	82.640	148000000	439438	340154	551373	2638	1398	1240
			(3219100)	(731455)	(4629052)			
Wage sum (real) p. c.	0.248	17.404	2.342	2.316	2.370	2506	1326	1180
in 1000 Euro, June			(0.877)	(0.910)	(0.846)			
Collective agr.	0	1	0.833	0.830	0.836	2638	1398	1240
			(0.373)	(0.375)	(0.370)			
Voluntary quits	0	95	2.689	2.413	ω	2638	1398	1240
			(4.589)	(4.495)	(4.676)			
Quit ratio	0	0.490	0.014	0.014	0.013	2638	1398	1240
			(0.023)	(0.025)	(0.021)			
Firings	1	230	10.236	9.381	11.202	2637	1398	1239
			(12.861)	(12.627)	(13.057)			
Profit situation (1-6)	1	9	3.260	3.281	3.235	289	153	136
			(1.099)	(1.127)	(1.070)			
Share of female workers	0	0.992	0.337	0.341	0.332	2632	1394	1238
			(0.253)	(0.256)	(0.250)			
Notes: IAB Establishment	Panel 1998-20	04. Variables valu	e added per cap	ita, quit ratio, sha	rre of female worke	rs, wage sur	n (real) f	rom own
calculations. Unly establish.	iments with a pe	rsistent works coun	cil since 1998.					

Table 2. Sample characteristics



Figure 3. Development of unconditional mean of voluntary quits in treated and untreated establishments 1998-2004

	(1) baseline	(2) voice	(3) baseline coll. agr.	(4) voice coll. agr.
Treat	0.626*	0.631*	0.667**	0.677*
	(0.367)	(0.378)	(0.368)	(0.378)
Post	$-1.796^{***}$	$-1.774^{***}$	-1.822***	$-1.796^{***}$
	(0.346)	(0.364)	(0.348)	(0.365)
Treat $\times$ Post	-0.888***	-0.923***	0.084	0.029***
	(0.301)	(0.303)	(0.519)	(0.532)
Coll. agr.	0.600	0.471	0.840**	0.717*
2	(0.382)	(0.394)	(0.397)	(0.412)
Treat $\times$ Post $\times$ Coll. agr.			$-1.137^{**}$	-1.120**
_			(0.510)	(0.519)
log(Sales)	-0.072	0.675	-0.083	2.681
	(0.432)	(12.252)	(0.433)	(12.277)
West	$-1.076^{***}$	-1.161***	$-1.036^{***}$	$-1.162^{***}$
	(0.374)	(0.406)	(0.373)	(0.406)
Employees	0.007	0.005	0.006	-0.007
	(0.007)	(0.063)	(0.007)	(0.063)
log(Value added p.c.)		-0.876		-2.903
		(12.542)		(12.564)
Wage sum p.c.		0.065		0.064
		(0.183)		(0.184)
Est. FE	yes	yes	yes	yes
Year FE	yes	yes	yes	yes
Industry dummys	yes	yes	yes	yes
$R^2$	0.061	0.061	0.063	0.063
Obs.	2638	2506	2638	2506
Establishments	1150	1090	1150	1090

Table 3. Regression results baseline model and model disentangling voice and monopoly	/
effects.	

Notes: Dependent variable is the number of voluntary quits. Standard errors clustered at the establishment level in parentheses. \*/ \*\*/ \*\*\* denote significance at the 10%/5%/1% level, respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (3) and (4) are 6.56 (0.002) and 6.82 (0.001), respectively. Wage sum per capita is measured in 1000 Euros.

	(5)	(6)	(7)	(8)
	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat	0.892	0.891	0.910	0.926
	(0.586)	(0.613)	(0.589)	(0.615)
Treat $\times$ 1998	0.082	0.203	-0.682	-0.180
	(0.676)	(0.711)	(2.082)	(2.187)
Treat $\times$ 1999	-0.629	-0.661	-0.257	-0.540
	(0.645)	(0.684)	(0.945)	(0.922)
Treat $\times$ 2000	-0.680	$-0.702^{-0.702}$	-0.433	$-0.530^{-1}$
	(0.622)	(0.654)	(1.135)	(1.160)
Treat $\times$ 2002	$-1.123^{**}$	$-1.183^{**}$	0.158	-0.105
	(0.487)	(0.502)	(0.656)	(0.670)
Treat $\times$ 2003	$-1.626^{***}$	$-1.662^{***}$	$-0.706^{-0.706}$	-0.771
	(0.605)	(0.638)	(0.895)	(0.938)
Treat $\times$ 2004	$-0.737^{'}$	-0.716	-0.284	$-0.108^{-0.108}$
	(0.629)	(0.652)	(0.945)	(0.994)
Treat × Coll.agr. × 1998	(0.020)	(0.00-)	0.795	0.385
			(2.048)	(2.149)
Treat $\times$ Coll agr $\times$ 1999			-0.441	-0.140
			(0.863)	(0.834)
Treat $\times$ Coll agr $\times$ 2000			(0.003) -0.287	(0.004)
ficat × Coll.agi. × 2000			(1.070)	(1.088)
Treat $\times$ Coll agr $\times$ 2002			(1.070)	(1.000) -1.968**
ffeat × Coll.agi. × 2002			(0.599)	(0.585)
Traat V Call agr V 2002			(0.388)	(0.365)
Theat $\times$ Coll.agr. $\times$ 2003			-1.044	-1.010
Treat v Call and v 2004			(0.824)	(0.850)
Ifeat $\times$ Coll.agr. $\times$ 2004			-0.520	-0.728
	0 504	0.450	(0.835)	(0.871)
Coll. agr.	0.594	0.456	0.834*	0.689
	(0.379)	(0.392)	(0.439)	(0.460)
log(Sales)	-0.056	0.646	-0.081	2.198
	(0.430)	(12.267)	(0.432)	(12.406)
West	-0.960***	-1.028**	-0.903**	-1.009**
	(0.355)	(0.412)	(0.358)	(0.421)
Employees	0.008	0.006	0.007	-0.004
	(0.007)	(0.064)	(0.008)	(0.064)
log(Value added p.c.)		-0.814		-2.390
		(12.530)		(12.666)
Wage sum p.c.		0.052		0.054
		(0.184)		(0.184)
Est. FE	yes	yes	yes	yes
Year FE	yes	yes	yes	yes
Industry dummys	yes	yes	yes	yes
$R^2$	0.065	0.065	0.067	0.067
Obs.	2638	2506	2638	2506
Establishments	1150	1090	1150	1090

Table 4. Regression results baseline model and model disentangling voice and monopoly effects with treatment effect, by collective agreement status and year.

Notes: Dependent variable is the number of voluntary quits. Standard errors clustered at the establishment level in parentheses. \*/ \*\*/ \*\*\* denote significance at the 10%/5%/1% level, respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (7) for the year 2000, 2002, 2003 and 2004 are 0.66 (0.516), 5.33 (0.005), 4.10 (0.017) and 0.86 (0.424), respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (6) for the year 2000, 2002, 2003 and 2004 are 0.60 (0.547), 4.64 (0.010), 3.87 (0.021) and 0.99 (0.373), respectively. Wage sum peg capita is measured in 1000 Euros.

# **Appendix: Robustness Checks**

Table 5. Regression results baseline model and model disentangling voice and monopoly effects, by collective agreement status. Sample of establishments with 150-195 and 205-250 employees.

	(1)	(2)	(3)	(4)
	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat	0.988	1.106*	1.036	1.173*
	(0.650)	(0.665)	(0.651)	(0.666)
Post	-1.727***	$-1.696^{***}$	$-1.746^{***}$	$-1.709^{***}$
	(0.389)	(0.407)	(0.389)	(0.408)
Treat $\times$ Post	-0.701*	-0.761 **	0.426	$0.442^{***}$
	(0.361)	(0.370)	(0.518)	(0.534)
Coll. agr.	0.657	0.547	0.968 * *	0.899 * *
	(0.406)	(0.425)	(0.424)	(0.448)
Treat $\times$ Post $\times$ Coll. agr.			$-1.362^{***}$	$-1.464^{***}$
			(0.516)	(0.534)
log(Sales)	-0.405	8.163	0.413	11.179
	(0.406)	(12.252)	(0.409)	(12.381)
West	-1.221***	$-1.425^{***}$	-1.173***	$-1.434^{***}$
	(0.409)	(0.439)	(0.409)	(0.439)
Employees	0.006	-0.038	0.004	-0.056
	(0.010)	(0.063)	(0.010)	(0.064)
log(Value added p.c.)		-8.826		-11.873
		(12.472)		(12.606)
Wage sum p.c.		-0.044		-0.042
		(0.205)		(0.206)
Est. FE	yes	yes	yes	yes
Year FE	yes	yes	yes	yes
Industry dummys	yes	yes	yes	yes
$R^2$	0.059	0.060	0.062	0.063
Obs.	2364	2243	2364	2243
Establishments	1100	1045	1100	1045

Notes: Dependent variable is the number of voluntary quits. The sample comprises establishments with 150-195 and 205-250 employees. Standard errors clustered at the establishment level in parentheses. \*/ \*\*/ \*\*\* denote significance at the 10%/5%/1% level, respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification 3 and 4 are 4.71 (0.009) and 5.20 (0.006), respectively. Wage sum per capita is measured in 1000 Euros.

	(5)	(6)	(7)	(8)
	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat	1.441*	1.560*	1.466*	1.637*
	(0.865)	(0.897)	(0.872)	(0.904)
Treat $\times$ 1998	-0.513	-0.378	-1.293	-0.751
	(0.755)	(0.797)	(2.291)	(2.400)
Treat $\times$ 1999	-1.099	-1.141	-0.824	-1.124
	(0.722)	(0.773)	(1.062)	(1.044)
Treat $\times$ 2000	-0.876	-0.912	-1.173	-1.324
	(0.657)	(0.696)	(1.210)	(1.232)
Treat $\times$ 2002	$-1.217^{**}$	$-1.286^{**}$	-0.085	-0.314
	(0.538)	(0.555)	(0.700)	(0.705)
Treat $\times$ 2003	$-1.538^{**}$	-1.599**	-0.447	-0.461
	(0.676)	(0.720)	(0.968)	(1.027)
Treat $\times$ 2004	-0.739	-0.791	0.041	0.385
	(0.727)	(0.762)	(1.057)	(1.121)
Treat $\times$ Coll.agr. $\times$ 1998			0.820	0.394
C			(2.241)	(2.348)
Treat $\times$ Coll.agr. $\times$ 1999			-0.331	0.008
C			(0.962)	(0.934)
Treat $\times$ Coll.agr. $\times$ 2000			0.365	0.509
C			(1.122)	(1.133)
Treat $\times$ Coll.agr. $\times$ 2002			$-1.376^{**}$	$-1.185^{*}$
C			(0.650)	(0.647)
Treat $\times$ Coll.agr. $\times$ 2003			-1.268	-1.332
6			(0.844)	(0.895)
Treat $\times$ Coll.agr. $\times$ 2004			-0.952	-1.465
6			(0.925)	(0.974)
Coll. agr.	0.652	0.539	0.889*	0.801
	(0.403)	(0.424)	(0.476)	(0.510)
log(Sales)	-0.414	7.871	-0.429	11.278
	(0.402)	(12.369)	(0.406)	(12.688)
West	$-1.039^{***}$	$-1.224^{***}$	-1.020**	-1.277***
	(0.394)	(0.450)	(0.401)	(0.461)
Employees	0.007	-0.035	0.006	-0.055
2	(0.010)	(0.064)	(0.010)	(0.066)
log(Value added p.c.)	(0.010)	-8.541	(01010)	-11.983
log( value added piel)		(12,556)		(12,883)
Wage sum p.c.		-0.067		-0.068
truge sum p.e.		(0.207)		(0.209)
Est FE	Ves	Ves	Ves	Ves
Year FE	ves	ves	Ves	Ves
Industry dummys	ves	ves	Ves	ves
$R^2$	0.063	0.064	0.066	0.067
Obs	2364	2243	2364	2243
Fotablichments	1100	1045	230 <del>4</del> 1100	1045
Establishinelits	1100	1045	1100	1045

Table 6. Regression results baseline model and model disentangling voice and monopoly effects, by collective agreement status and year. Sample of establishments with 150-195 and 205-250 employees.

Notes: Dependent variable is the number of voluntary quits. The sample comprises establishments with 150-195 and 205-250 employees. Standard errors clustered at the establishment level in parentheses. \*/ \*\*/ \*\*\* denote significance at the 10%/5%/1% level, respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (7) for the year 2000, 2002, 2003 and 2004 are 0.90 (0.407), 4.19 (0.015), 3.57 (0.029) and 1.06 (0.346), respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post × Coll. Agr.) in specification (8) for the year 2000, 2002, 2003 and 2004 are 0.90 (0.23), 3.43 (0.033) and 1.76 (0.173), respectively. Wage sum per capita is measured in 1000 Euros.

	(1)	(2)	(3)	(4)
	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat	1.151*	1.253*	1.198*	1.322*
	(0.683)	(0.709)	(0.685)	(0.711)
Post	$-1.493^{***}$	$-1.426^{***}$	-1.507***	$-1.437^{***}$
	(0.402)	(0.427)	(0.402)	(0.427)
Treat $\times$ Post	-0.840 **	-0.980 **	0.307	0.175
	(0.421)	(0.433)	(0.598)	(0.612)
Coll. agr.	0.762*	0.674	1.062**	0.994 **
	(0.447)	(0.481)	(0.463)	(0.504)
Treat $\times$ Post $\times$ Coll. agr.			$-1.363^{**}$	$-1.383^{**}$
			(0.581)	(0.587)
log(Sales)	-0.045	7.059	-0.055	9.492
	(0.369)	(13.937)	(0.375)	(14.030)
West	-1.097 **	$-1.296^{***}$	-1.049***	-1.297 ***
	(0.435)	(0.483)	(0.434)	(0.484)
Employees	0.000	-0.037	-0.002	-0.051
	(0.010)	(0.073)	(0.010)	(0.074)
log(Value added p.c.)		-7.226		-9.692
		(14.147)		(14.244)
Wage sum p.c.		0.148		0.147
		(0.218)		(0.219)
Est. FE	yes	yes	yes	yes
Year FE	yes	yes	yes	yes
Industry dummys	yes	yes	yes	yes
$R^2$	0.071	0.072	0.074	0.075
Obs.	2087	1978	2087	1978
Establishments	1030	981	1030	981

Table 7. Regression results baseline model and model disentangling voice and monopoly effects, by collective agreement status. Sample of establishments with 150-190 and 210-250 employees.

Notes: Dependent variable is the number of voluntary quits. The sample comprises establishments with 150-190 and 210-250 employees. Standard errors clustered at the establishment level in parentheses. \*/ \*\*/ \*\*\* denote significance at the 10%/5%/1% level, respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (3) and (4) are 4.25 (0.015) and 4.74 (0.009), respectively. Wage sum per capita is measured in 1000 Euros.

	(5)	(6)	(7)	(8)
	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat	1.461*	1.565*	1.482*	1.648*
	(0.847)	(0.891)	(0.859)	(0.908)
Treat $\times$ 1998	-0.422	-0.267	-1.073	-0.527
	(0.789)	(0.829)	(2.408)	(2.407)
Treat $\times$ 1999	-0.775	-0.801	-0.307	-0.675
	(0.771)	(0.838)	(1.203)	(1.176)
Treat $\times$ 2000	-0.482	-0.526	-0.644	-0.807
	(0.684)	(0.725)	(1.478)	(1.510)
Treat $\times$ 2002	$-1.116^{*}$	-1.278**	0.178	-0.232
	(0.599)	(0.618)	(0.757)	(0.749)
Treat $\times$ 2003	$-1.331^{*}$	$-1.429^{*}$	-0.268	-0.408
	(0.692)	(0.733)	(1.132)	(1.132)
Treat $\times$ 2004	$-0.990^{-1}$	$-1.165^{'}$	-0.165	0.012
	(0.693)	(0.736)	(1.167)	(1.232)
Treat $\times$ Coll.agr. $\times$ 1998	()		0.708	0.271
			(2.241)	(2.357)
Treat $\times$ Coll.agr. $\times$ 1999			-0.552	-0.141
			(1.109)	(1.080)
Treat $\times$ Coll agr $\times$ 2000			0.177	0.319
			(1.401)	(1.421)
Treat $\times$ Coll agr $\times$ 2002			-1 550**	-1.255*
			(0.691)	(0.673)
Treat $\times$ Coll agr $\times$ 2003			(0.031) -1.234	(0.019) -1 199
			(0.969)	(1.033)
Treat $\times$ Coll agr $\times$ 2004			(0.903) -1.001	(1.000)
ficat × Coll.agi. × 2004			(1.001)	$(1 \ 104)$
Coll agr	0 769*	0.668	(1.030) 1.030**	(1.104) 0.053*
Coll. agi.	(0.448)	(0.481)	(0.523)	(0.555)
log(Sales)	0.440)	(0.401)	(0.525)	(0.572)
log(Sales)	(0.373)	(14.201)	(0.370)	(14,600)
West	(0.373) -1.038**	(14.291) -1 915**	(0.575) -1.004**	(14.033) -1.230**
west	(0.428)	(0.508)	(0.425)	(0.521)
Employees	(0.428)	(0.308)	(0.433)	(0.521)
Employees	(0.001)	-0.030	-0.001	-0.032
$\log(V_{\rm obs})$	(0.010)	(0.075)	(0.010)	(0.078) 10.170
log(value added p.c.)		-7.410		-10.179
Waga aum n a		(14.404)		(14.699)
wage sum p.c.		(0.131)		(0.120)
		(0.220)		(0.222)
ESI. FE	yes	yes	yes	yes
Ical PE	yes	yes	yes	yes
Industry dummys	yes	yes	yes	yes
К" 01	0.073	0.074	0.076	0.077
Ubs.	2087	1978	2087	1978
Establishments	1030	981	1030	981

Table 8. Regression results baseline model and model disentangling voice and monopoly effects, by collective agreement status and year. Sample of establishments with 150-190 and 210-250 employees.

Notes: Dependent variable is the number of voluntary quits. The sample comprises establishments with 150-190 and 210-250 employees. Standard errors clustered at the establishment level in parentheses. \*/ \*\*/ \*\*\* denote significance at the 10%/5%/1% level, respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (7) for the year 2000, 2002, 2003 and 2004 are 0.26 (0.768), 3.64 (0.027), 2.58 (0.076) and 1.71 (0.181), respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post × Coll. Agr.) in specification (8) for the year 2000, 2002, 2003 and 2004 are 0.28 (0.763), 3.23 (0.040), 2.52 (0.081) and 2.48 (0.085), respectively. Wage sum per capita is measured in 1000 Euros.

	(1)	(2)	(3)	(4)
	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat	0.377	0.383	0.410	0.422
	(0.351)	(0.365)	(0.349)	(0.364)
Post	$-1.713^{***}$	-1.659 ***	$-1.743^{***}$	$-1.689^{***}$
	(0.382)	(0.397)	(0.384)	(0.399)
Treat $\times$ Post	-0.257	-0.293	0.910	0.820
	(0.373)	(0.394)	(0.718)	(0.732)
Coll. agr.	0.606	0.476	0.979 * *	0.848*
	(0.386)	(0.398)	(0.429)	(0.450)
Treat $\times$ Post $\times$ Coll. agr.			-1.350*	-1.298*
			(0.695)	(0.718)
log(Sales)	-0.115	-1.178	-0.111	0.553
	(0.429)	(12.402)	(0.428)	(12.667)
West	-1.300***	-1.358***	-1.261***	-1.353
	(0.373)	(0.395)	(0.375)	(0.394)
Employees	0.007	0.014	0.005	0.003
	(0.007)	(0.064)	(0.007)	(0.066)
log(Value added p.c.)		0.950		-0.787
		(12.683)		(12.941)
Wage sum p.c.		0.065		0.064
		(0.185)		(0.185)
Est. FE	yes	yes	yes	yes
Year FE	yes	yes	yes	yes
Industry dummys	yes	yes	yes	yes
$R^2$	0.058	0.057	0.061	0.060
Obs.	2638	2506	2638	2506
Establishments	1150	1090	1150	1090

Table 9.	Regression results baseline model and n	odel disentang	ling voice and m	onopoly
effects,	by collective agreement status. Placebo	nalysis for the	year 2000.	

Notes: Dependent variable is the number of voluntary quits. Standard errors clustered at the establishment level in parentheses. \*/\*\*/\*\*\* denote significance at the 10%/5%/1% level, respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (3) and (4) are 2.17 (0.114) and 1.92 (0.147), respectively. For the placebo analysis, the year of treatment has been predated to the year 2000. Wage sum per capita is measured in 1000 Euros.

# Table 10. Treatment effects for baseline model and model disentangling voice and monopoly effects, comparison of OLS and Poisson coefficients

		(1)	(2)	(3)	(4)
	Estimated model	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat $\times$ Post	OLS	-0.888***	$-0.923^{***}$	0.084	0.029
	Poisson	$-0.362^{***}$	$-0.365^{***}$	0.078	0.076
Treat $\times$ Post $\times$ Coll. agr.	OLS			-1.137**	-1.120 **
	Poisson			-0.528***	$-0.531^{***}$

Notes: Dependent variable is the number of voluntary quits. Standard errors are robust. \*/ \*\*/ \*\*\* denote significance at the 10%/5%/1% level, respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (3) for OLS and Poisson are 6.56 (0.002) and 14.45 (0.001), respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post × Coll. Agr.) in specification (4) for OLS and Poisson are 6.56 (0.002) and 20.08 (0.000), respectively.

		(5)	(6)	(7)	(8)
	Estimated model	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat $\times$ 1998	OLS	0.082	0.203	-0.682	-0.180
	Poisson	0.257	0.344	0.285	0.372
Treat $\times$ 1999	OLS	-0.629	-0.661	-0.257	-0.540
	Poisson	-0.128	-0.130	0.065	-0.035
Treat $\times$ 2000	OLS	-0.680	-0.702	-0.433	-0.530
	Poisson	-0.171	-0.181	-0.151	-0.181
Treat $\times$ 2002	OLS	$-1.123^{**}$	$-1.183^{**}$	0.158	-0.105
	Poisson	-0.344 **	$-0.356^{**}$	0.101	0.053
Treat $\times$ 2003	OLS	$-1.626^{***}$	$-1.662^{***}$	-0.706	-0.771
	Poisson	$-0.596^{***}$	$-0.589^{**}$	-0.110	-0.104
Treat $\times$ 2004	OLS	-0.737	-0.716	0.084	-0.108
	Poisson	-0.278	-0.269	0.022	0.090
Treat $\times$ Post $\times$ Coll. agr. $\times$ 1998	OLS			0.795	0.385
	Poisson			-0.034	-0.044
Treat $\times$ Post $\times$ Coll. agr. $\times$ 1999	OLS			-0.442	-0.140
	Poisson			-0.232	-0.112
Treat $\times$ Post $\times$ Coll. agr. $\times$ 2000	OLS			-0.287	-0.196
	Poisson			-0.016	0.005
Treat $\times$ Post $\times$ Coll. agr. $\times$ 2002	OLS			-1.510**	-1.268**
	Poisson			-0.541***	-0.499 ***
Treat $\times$ Post $\times$ Coll. agr. $\times$ 2003	OLS			-1.044	-1.016
	Poisson			$-0.555^{***}$	-0.552 **
Treat $\times$ Post $\times$ Coll. agr. $\times$ 2004	OLS			-0.520	-0.728
	Poisson			-0.368	-0.456

# Table 11. Treatment effects for baseline model and model disentangling voice and monopoly effects, comparison of OLS and poisson coefficients

Notes: Dependent variable is the number of voluntary quits. Standard errors are robust. \*/ \*\*\*/ \*\*\*\* denote significance at the 10%/5%/1% level, respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (7) for the regression for the year 2000, 2002, 2003 and 2004 are 0.66 (0.516), 5.33 (0.005), 4.10 (0.017) and 0.86 (0.424), respectively. Statistics and corresponding p-values from  $\chi^2$ -test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (7) for the poisson regression for the year 2000, 2002, 2003 and 2004 are 0.86 (0.424), respectively. Statistics and corresponding p-values from  $\chi^2$ -test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (7) for the poisson regression for the year 2000, 2002, 2003 and 2004 are 0.81 (0.666), 12.56 (0.002), 9.41 (0.001) and 2.59 (0.274), respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (8) for the OLS regression for the year 2000, 2002, 2003 and 2004 are 0.60 (0.547), 4.64 (0.010), 3.87 (0.021) and 0.99 (0.373), respectively. Statistics and corresponding p-values from  $\chi^2$ -test on joint significance of (Treat × Post) × Coll. Agr.) in specification (8) for the poisson regressions for the year 2000, 2002, 2003 and 2004 are 0.60 (0.547), 4.64 (0.010), 3.87 (0.021) and 0.99 (0.373), respectively. Statistics and corresponding p-values from  $\chi^2$ -test on joint significance of (Treat × Post) × Coll. Agr.) in specification (8) for the poisson regressions for the year 2000, 2002, 2003 and 2004 are 0.86 (0.650), 12.08 (0.002), 8.12 (0.017) and 3.23 (0.198), respectively.

	(1)	(2)	(3)	(4)
	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat	0.0026	0.0026	0.0028	0.0028
	(0.0018)	(0.0019)	(0.0018)	(0.0019)
Post	-0.0093***	-0.0092***	-0.0094***	-0.0093***
	(0.0018)	(0.0019)	(0.0018)	(0.0019)
Treat $\times$ Post	-0.0033*	$-0.0034^{**}$	-0.0016	-0.0014
	(0.0015)	(0.0015)	(0.0026)	(0.0026)
Coll. agr.	0.0033*	0.0026	$0.0045^{**}$	0.0038*
	(0.0020)	(0.0020)	(0.0021)	(0.0021)
Treat $\times$ Post $\times$ Coll. agr.			-0.0057**	$-0.0056^{**}$
			(0.0024)	(0.0025)
log(Sales)	-0.0002	-0.0166	-0.0002	-0.0066
	(0.0023)	(0.0638)	(0.0023)	(0.0637)
West	-0.0026	-0.0025	-0.0024	-0.0025
	(0.0019)	(0.0022)	(0.0019)	(0.0022)
Employees	0.0000	-0.0001	0.0000	0.0000
	(0.0000)	(0.0003)	(0.0000)	(0.0003)
log(Value added p.c.)		0.0160		0.0059
		(0.0656)		(0.0655)
Wage sum p.c.		0.0003		0.0003
		(0.0010)		(0.0010)
Est. FE	yes	yes	yes	yes
Year FE	yes	yes	yes	yes
Industry dummys	yes	yes	yes	yes
$R^2$	0.050	0.048	0.051	0.050
Obs.	2638	2506	2638	2506
Establishments	1150	1090	1150	1090

Table 12. Dependent variable quit ratio: Regression results baseline model and model disentangling voice and monopoly effects.

Notes: Dependent variable is the quit ratio. Standard errors clustered at the establishment level in parentheses. \*/ \*\*/ \*\*\* denote significance at the 10%/5%/1% level, respectively. Coefficients and standard errors are rounded at the fourth decimal. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (3) and (4) are 5.08 (0.006) and 5.08 (0.006), respectively. Wage sum per capita is measured in 1000 Euros.

	(5)	(6)	(7)	(8)
Spec.	baseline	voice	baseline coll. agr.	voice coll. agr.
Treat	0.0033	0.0032	0.0034	0.0033
	(0.0028)	(0.0029)	(0.0028)	(0.0030)
Treat $\times$ 1998	0.0022	0.0029	-0.0009	0.0014
	(0.0032)	(0.0034)	(0.0097)	(0.0102)
Treat $\times$ 1999	-0.0020	-0.0021	0.0003	-0.0009
	(0.0031)	(0.0033)	(0.0044)	(0.0044)
Treat $\times$ 2000	-0.0031	-0.0031	-0.0014	-0.0018
	(0.0031)	(0.0032)	(0.0056)	(0.0056)
Treat $\times$ 2002	-0.0043*	-0.0045*	0.0018	0.0007
	(0.0023)	(0.0024)	(0.0031)	(0.0032)
Treat $\times$ 2003	-0.0063 **	-0.0064 **	-0.0012	-0.0015
	(0.0031)	(0.0032)	(0.0044)	(0.0046)
Treat $\times$ 2004	-0.0013	-0.0011	-0.0013	0.0021
	(0.0032)	(0.0033)	(0.0047)	(0.0047)
Treat $\times$ Coll.agr. $\times$ 1998			0.0031	0.0014
			(0.0095)	(0.0100)
Treat $\times$ Coll.agr. $\times$ 1999			-0.0028	-0.0014
			(0.0040)	(0.0039)
Treat $\times$ Coll.agr. $\times$ 2000			-0.0020	-0.0015
			(0.0051)	(0.0052)
Treat $\times$ Coll.agr. $\times$ 2002			$-0.0072^{***}$	-0.0061**
			(0.0027)	(0.0027)
Treat $\times$ Coll.agr. $\times$ 2003			-0.0058	-0.0056
			(0.0040)	(0.0041)
Treat $\times$ Coll.agr. $\times$ 2004			-0.0032	-0.0039
			(0.0038)	(0.0040)
Coll. agr.	0.0032*	0.0025	$0.0046^{**}$	0.0038
	(0.0020)	(0.0020)	(0.0023)	(0.0024)
log(Sales)	-0.0001	-0.0168	-0.0002	-0.0091
	(0.0023)	(0.0636)	(0.0023)	(0.0639)
West	-0.0018	-0.0017	-0.0015	-0.0015
	(0.0019)	(0.0023)	(0.0019)	(0.0023)
Employees	0.0000	0.0001	0.0000	0.0000
	(0.0000)	(0.0003)	(0.0000)	(0.0003)
log(Value added p.c.)		0.0164		0.0086
		(0.0653)		(0.0656)
Wage sum p.c.		0.0003		0.0003
		(0.0010)		(0.0010)
Est. FE	yes	yes	yes	yes
Year FE	yes	yes	yes	yes
Industry dummys	yes	yes	yes	yes
$R^2$	0.053	0.053	0.056	0.054
Obs.	2638	2506	2638	2506
Establishments	1150	1090	1150	1090

Table 13. Dependent variable quit ratio: Regression results baseline model and model disentangling voice and monopoly effects with treatment effect, by collective agreement status and year.

Notes: Dependent variable is the quit ratio. Standard errors clustered at the establishment level in parentheses. \*/ \*\*/ \*\*\* denote significance at the 10%/5%/1% level, respectively. Coefficients and standard errors are rounded at the fourth decimal. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post) and (Treat × Post × Coll. Agr.) in specification (7) for the year 2000, 2002, 2003 and 2004 are 0.59 (0.553), 4.83 (0.008), 2.80 (0.061) and 0.42 (0.657), respectively. Statistics and corresponding p-values from F-test on joint significance of (Treat × Post × Coll. Agr.) in specification (8) for the year 2000, 2002, 2003 and 2004 are 0.51 (0.602), 3.92 (0.020), 2.64 (0.072) and 0.54 (0.583), respectively. Wage sum per capita is measured in 1000 Euros.